Author's response to reviews

Title: Study protocol for the 10 Top Tips (10TT) Trial: Randomised controlled trial of habit-based advice for weight control in general practice

Authors:

Rebecca J Beeken (r.beeken@ucl.ac.uk)
Helen Croker (h.croker@ucl.ac.uk)
Stephen Morris (steve.morris@ucl.ac.uk)
Baptiste Leurent (b.leurent@ucl.ac.uk)
Rumana Omar (rumana@stats.ucl.ac.uk)
Irwin Nazareth (i.nazareth@ucl.ac.uk)
Jane Wardle (j.wardle@ucl.ac.uk)

Version: 2 Date: 23 July 2012

Author's response to reviews: see over
23 July 2012

Dear Mr Dizon

Re: MS 1805917629765685; Study protocol for the 10 Top Tips (10TT) Trial: Randomised controlled trial of habit-based advice for weight control in general practice

Thank you for your e-mail dated 23/07/2012 concerning the changes required for the above manuscript to be sent out for peer review. As requested, we are re-submitting the manuscript with the following changes:

1. We have added the TRN at the end of the Abstract in the submissions system.
2. We have added a competing interests section.
3. We have included an acknowledgments section.

In addition to the requested changes, we have also added a List of abbreviations and Authors contributions section, as these were also missing from the original manuscript.

We hope these changes meet with your satisfaction and look forward to hearing your views.

Yours sincerely

Jane Wardle